

This is a repository copy of Understanding the relationship between social determinants of health and maternal mortality.

White Rose Research Online URL for this paper: <u>https://eprints.whiterose.ac.uk/183811/</u>

Version: Accepted Version

Article:

Jones, G.L., Mitchell, C.A. orcid.org/0000-0002-4790-0095, Hirst, J.E. et al. (1 more author) (2022) Understanding the relationship between social determinants of health and maternal mortality. BJOG: An International Journal of Obstetrics & Gynaecology, 129 (7). pp. 1211-1228. ISSN 1470-0328

https://doi.org/10.1111/1471-0528.17044

This is the peer reviewed version of the following article: Jones, GL, Mitchell, CA, Hirst, JE, Anumba, DOC; on behalf of the Royal College of Obstetricians and Gynaecologists. Understanding the relationship between social determinants of health and maternal mortality. BJOG: Int J Obstet Gy. 2022; 00: 1– 18., which has been published in final form at https://doi.org/10.1111/1471-0528.17044. This article may be used for non-commercial purposes in accordance with Wiley Terms and Conditions for Use of Self-Archived Versions. This article may not be enhanced, enriched or otherwise transformed into a derivative work, without express permission from Wiley or by statutory rights under applicable legislation. Copyright notices must not be removed, obscured or modified. The article must be linked to Wiley's version of record on Wiley Online Library and any embedding, framing or otherwise making available the article or pages thereof by third parties from platforms, services and websites other than Wiley Online Library must be **Revisibited**.

Items deposited in White Rose Research Online are protected by copyright, with all rights reserved unless indicated otherwise. They may be downloaded and/or printed for private study, or other acts as permitted by national copyright laws. The publisher or other rights holders may allow further reproduction and re-use of the full text version. This is indicated by the licence information on the White Rose Research Online record for the item.

Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.



Scientific Impact Paper No. XX Fourth Draft – July 2021 (v4.4)

New Title: Understanding the relationship between social determinants of health and maternal mortality.

Note from RCOG Guidance Editorial Manager: Please ignore the formatting of the references for now

GL Jones, C Mitchell, J Hirst, DOC Anumba, on behalf of the Royal College of Obstetricians and Gynaecologists

Correspondence: Royal College of Obstetricians and Gynaecologists, 27 Sussex Place, Regent's Park, London NW1 4RG.

Email: clinicaleffectiveness@rcog.org.uk

Plain language summary

This paper aims to inform those involved in the care of pregnant women in the UK about the relationship between social determinants of health and the risk of maternal death. Social determinants are the social conditions in which people are born, grow, live, work and age. These conditions can shape a person's health, their exposure to illness and access to care. The worse social and economic positions people are in, the worse their health outcomes are likely to be.

Exploring these issues in relation to maternal outcomes is timely because the profile of mothers giving birth in the UK is changing. Increasingly, babies are being born to older or obese mothers, or to women born outside of the UK.

The evidence suggests a strong relationship between social determinants and poor maternal outcomes, including an increased risk of maternal death. Despite the provision of free reproductive and maternity services, current models of care are still failing those pregnant women that have lived in adverse social circumstances prior to, during and after pregnancy. The reasons are complex but maternal outcomes are particularly poor for socially disadvantaged women affected by pre-existing physical or mental health problems, those who misuse substances, have a low level of education, are overweight, undernourished or poorly sheltered, and those who are at increased risk due to the threat of abusive and unsupportive partners, families and peers. Some traditional, cultural and social norms, and reproductive laws and policies, also appear to influence a pregnant woman's relationships, behaviour and access to healthcare services. These factors can reinforce a web of social disadvantage and poor outcomes for some pregnant women, which can persist for future generations. These issues have been brought into public awareness as a result of the Covid-19 pandemic which has highlighted the ways that health inequalities can adversely affect health outcomes beyond an individual's control. However, addressing the relationship between Covid-19 and social determinants of health is outside the scope of this paper. Whilst social determinants of maternal health also have an impact on offspring health, this is outside the remit of this paper too.

Healthcare professionals have limited power on their own to modify social determinants of health, but this paper concludes with current recommendations to improve the maternal outcomes of vulnerable pregnant women residing in the UK and proposes improvements to the collection and reporting of maternal outcome data, with further investment in prevention and intervention measures to better support these women.

1. Background

This paper aims to improve understanding concerning the relationship between social determinants of health and maternal outcomes. Social determinants are the factors in the social environment that can shape a person's health, their propensity to illness and access to care. They are the social conditions into which people are born, grow, live, work and age¹ that contribute or detract from them being able to live a healthy life. Understanding the social determinants of health is a global and national priority. In March 2005 the WHO launched The Commission on Social Determinants of Health (CSDH) to support countries and global health partners in addressing the social injustice of health inequalities.² The evidence suggested that there is a clear social gradient, i.e. the more disadvantaged people are in terms of their social and economic position within society, the worse their health outcomes in terms of morbidity (suffering from a disease or condition) and mortality (death), which can persist through future generations and influence behaviour beyond an individual's control.^{1,2} This is not a new phenomenon. Historical data has demonstrated the links between the social position of the poorest 'working classes' generally and their health outcomes.³ The Marmot Review was commissioned by the British Government to identify how the WHO 2005 findings could be specifically applied to develop a strategy to reduce health inequalities in England, the outcome of which was 'Fair Society, Healthy Lives'.¹ This report acknowledged that the magnitude of health inequalities in England was still considerable, although less than in many other parts of the world. Overall, people from poorer areas die sooner and also live more of their shorter lives in worse health.

In the UK, over the last five three-year periods as reported by MBRRACE-UK, there has not been any significant change in the overall maternal death rate (including both direct and indirect causes of death).⁴ In addition, since 2003 in each reported three-year period, more of these maternal deaths are as a result of indirect rather than direct causes. In the latest report at the time of writing (covering the years 2016–18 and published in December 2020), indirect causes account for 58% of maternal deaths.⁴ Pre-existing medical and mental health comorbidities and factors contributing to health inequalities, such as obesity, smoking, substance misuse, poor utilisation of antenatal care, maternal age, unemployment and ethnicity, among others, have been identified as important risk factors associated with direct and indirect causes of maternal mortality.^{5–8}.

Whilst social determinants have been linked to both maternal morbidity and mortality, most of the evidence to date has focused upon maternal mortality which is at the end of the spectrum of all possible health impacts and is therefore the focus of this paper. This paper also primarily focuses on the UK and therefore draws upon data from the Mothers and Babies: Reducing Risk through Audits and Confidential Enquiries across the UK (MBRRACE-UK) reports and the official National Statistics from the UK's Office for National Statistics (ONS). However, where UK evidence is lacking, it will also consider other national and international evidence, as defined by those countries in the Organisation for Economic Co-operation and Development (OECD) where healthcare is free at the point of delivery.

Finally, it is worth highlighting that the provision of maternity care in the UK is based upon a risk model of assessment, whereby the risk factors identified for a possible adverse pregnancy outcome are used to determine the care the pregnant woman receives. The complex causal pathways linking social determinants and maternal death are extremely difficult to disentangle with the current levels of evidence available. However, what is clear is that the risk factors associated with a poor maternal outcome presented below are typically outside of the woman's control and are often indicators of this social inequality and disadvantage.

2. Defining maternal outcomes and social determinants

Maternal deaths are defined by the WHO⁹ and adopted by MBBRACE-UK⁴ as follows:

- Maternal death refers to the death of a woman while pregnant or within 42 days (6 weeks) of termination of pregnancy (termination refers to the end of a pregnancy, irrespective of the duration and the site of the pregnancy, from any cause related to or aggravated by the pregnancy or its management, but not from accidental or incidental causes). This is further subdivided into two groups:
 - direct obstetric deaths: those deaths resulting from obstetric complications of the pregnancy state (pregnancy, labour and the puerperium), from interventions, omissions, incorrect treatment, or from a chain of events resulting from any of the above.
 - *indirect obstetric deaths*: those deaths resulting from previous existing disease that developed during pregnancy and which was not due to direct obstetric causes, but which was aggravated by physiological effects of pregnancy.
- Late maternal death refers to the death of a woman after 42 days (6 weeks) but less than 1 year post termination of pregnancy.

The World Health Organization (WHO) defines social determinants as the factors which can mediate an individual's living conditions, psycho-social circumstances, behavioural and biological factors and also access to health care. These factors can be at an individual (such as age, gender, pre-existing health conditions, family and interpersonal relationships), institutional or structural level [such as income, housing, occupation, education, environment, culture, political) and legal (migration and discrimination)]. Understanding the social determinants of health is a global and national priority as health outcomes are closely linked to living conditions, the quality of the social environment and the inequitable distribution of wealth, power and resources.²

Several frameworks have been proposed by the WHO to address the relationships between social determinants and health outcomes.¹⁰⁻¹¹ In 2011, a framework was proposed specifically in relation to maternal health (Appendix I).¹² We have used this framework to conceptualise how social determinants can contribute to maternal mortality in the UK. The framework outlines the intermediary and structural factors which may impact upon maternal health. It also places mothers within their families and social networks, both of which are considered the levels at which poor maternal outcomes are directly experienced.¹² However, for the purposes of this paper, the framework has been simplified to provide a model that considers risk factors and institutional and structural factors, which better fits with the risk model of assessment and the UK setting of maternal healthcare delivery. Also, in practice there is typically limited data upon which to provide a rigorous and robust analysis of all of these intermediary and structural factors and their relationship to maternal death, but the existing literature has been synthesised and appraised where possible.

3. Risk factors for adverse pregnancy outcomes and their links to social determinants

The following risk factors for possible maternal mortality are discussed below in the context of their links to social determinants.

3.1 Maternal age

The average age of mothers at childbirth in 2018 was 30.6 years, an increase of 4 years over the previous four decades.¹³ The evidence suggests that mothers aged 35 years and older have a high

relative risk of maternal mortality. This was especially pronounced in mothers aged over 40 years, whereby the relative risk of maternal death was highest when compared with those aged 20–24 years during the MBRRACE-UK trienniums 2016–18 (RR 4.65; CI 2.48–8.89) and 2012–14 (RR 3.00; CI 1.51–5.83).⁴ This is important since conception rates in women aged 40 years and older more than doubled between 1991 and 2018.¹⁴

Factors related to maternal aging, including a higher prevalence of pre-existing medical conditions and utilisation of assisted reproductive technologies, are associated with increased risk of maternal mortality. Yet the socio-economic position of older mothers is often better than in younger mothers.¹⁵ Some of the reasons why women postpone childbearing until they are older include time spent in education to further career opportunities, workplace inflexibility and establishing financial independence.¹⁶ However, the association between age, social inequality and maternal mortality is more complicated. In women over 35 years of age, smoking, pre-existing medical conditions, older maternal age, previous pregnancy problems and less use of antenatal care have all been associated with increased mortality.¹⁷ Thus, whilst older mothers overall may be in relatively better social positions compared to younger mothers, social factors remain important drivers of maternal mortality in this group.

While young age alone is not a risk factor for maternal mortality in England and Wales, teenage pregnancies indicate a strong social gradient. There continues to be a welcome decrease in the conception rate for women aged under 18 years old.¹⁴ However, young women from the poorest backgrounds are more likely to become teenage mothers than young women from affluent backgrounds. For example, in 2018, the conception rate for women under 18 years was higher in the 50% most deprived areas in England.¹⁴ Conversely, the percentage of conceptions leading to a legal abortion was higher in the 50% least deprived areas in England. Teenage mothers are more likely to leave education, be unemployed, live in the most deprived areas and in poverty and be single parents compared with older mothers.^{15,18} However, some research suggests that teenage parenting within marriage may not be perceived as a problem, or a distinctive event, particularly for young parents of Muslim faith.¹⁹ Whilst the ONS does not report conception rates by ethnicity, a report in 2013, found mothers aged under 20 years old from Black Caribbean and White British backgrounds accounted for the highest percentage of teenagers giving birth in England and Wales.²⁰

3.2 Parity

Parity refers to the number of times that a woman has given birth to a fetus with a gestational age of 24 weeks or more, regardless of whether the child was born alive or was stillborn. High parity is an important factor driving maternal mortality globally and is associated with lower socio- economic standing, but in developed countries such as the UK where family sizes are smaller, this is arguably less of a major contributor to mortality. The highest proportion of mothers who died from all causes between 2016–18 had one or two previous children (44%)⁴ with 36% in their first pregnancy, a finding that has been relatively consistent since 2011.^{4,21,22} This is to be expected as the average number of children born to mothers in the UK is 1.9. This, combined with the low absolute numbers of women who died with high parity, it is not possible to unpick the relationship between social determinants and this finding for UK data. Evidence suggests that Black, Asian and minority ethnic women are more likely to be multiparous and therefore more prone to obstetric haemorrhage, and worse maternal outcomes as a consequence.²³⁻²⁵ However, in the UK, generally multiparity (having given birth two or more times) was recently shown not to be associated with death in women over 35-years of age.¹⁷

3.3. Pre-existing health conditions

Preconception health is important for subsequent maternal and child health. The results of a national unmatched case-control analysis using data from two sources, compared women who died from both direct and indirect causes from 2009–13 and women who did not have any life-threatening complications during pregnancy and childbirth. Case data was obtained from the Confidential Enquiry into Maternal Deaths and control data selected from the UK Obstetric Surveillance System (UKOSS). They reported an adjusted odds ratio of 8.65 (95% CI 6.29–11.90). This revealed that the odds of maternal mortality from direct or indirect causes was nearly nine-fold higher among women who had a pre-existing physical and mental health condition compared with women who did not.⁸

The latest MBRRACE-UK 2016-2018 report identified a pre-existing medical problem (excluding obesity) in 66%, of women who died during pregnancy or in the 6 weeks after birth (section 4.1 below) and 35% had a pre-existing mental health problem or psychiatric disorder (section 4.3 below).⁴ Cardiac disease remains the leading cause of indirect death, and the rate has not significantly changed over the last fifteen years. However, the relationships that may exist between these conditions and the social determinants of health are complex and remain unclear.

Integration of complex medical and antenatal care may be influenced by the social circumstances of women. Undiagnosed and sub-optimally treated medical conditions may be a cause of maternal morbidity and mortality. The ability for women to access healthcare, including routine antenatal care to identify complications of pregnancy, adhere to medication and self-manage long-term conditions, such as epilepsy and diabetes mellitus, are all influenced by social determinants.²⁶ In the triennium 2015–17, *'severe and multiple disadvantage'* whereby women had three or more of the following indicators of social disadvantage: substance abuse, domestic abuse, abuse in childhood, arrival in UK within last 5 years, refugee or asylum seeker, mental health diagnosis, female genital mutilation, and known learning difficulties, was highlighted as a major concern in 6% of all deaths overall.²⁷ However, this has increased to 8% in the most recent MBRRACE-UK 2016-2018 report.⁴ These complexities can be observed in the case histories of some of the women. For example, *'a pregnant woman who was an asylum seeker, the sole carer for other children and had hypertension and obesity booked late for antenatal care'*. Biopsychosocial complexity and service factors contributed to poor adherence to medications, irregular monitoring of her blood pressure and she died of hypertensive heart disease.²⁷

3.4 Non communicable disease

Cardiac disease continues to be the leading cause of maternal mortality in the UK a rate which has remained unchanged for more than 15 years. Almost a quarter (23%) of maternal deaths were related to cardiac conditions, 16% to venous thromboembolism (VTE), with neurological conditions such as stroke or epilepsy contributing 13% of maternal deaths between 2015 and 2017.²⁷ From the 2020 report, there was a statistically non-significant increase in mortality rates from neurological causes with a significant increase in maternal morality due to Sudden Unexpected Death in Epilepsy (SUDEP) with a more than doubling between 2013-2015 and 2016-2018 (RR 2.33, 95% 0.96-6.19, p=0.04). Identified areas for action include ensuring that pregnant women with epilepsy are facilitated to access specialist services early in pregnancy and ideally receive pre-pregnancy counselling.⁴

Women who are older, more obese, smoke, who have family history of ischaemic heart disease or comorbid diabetes mellitus or hypertension are at overall higher risk of cardiovascular disease and of ischaemic heart disease (the leading cardiac cause of maternal mortality).²⁷ In the 2015–17 triennium, the majority of women who died from a cardiac cause were over 30 years old (64%) and had previous (unspecified) pre-existing 'health problems' (62%). Of the women who died, 73% were not known to have pre-existing cardiac problems, raising concerns that opportunities to diagnose cardiac problems earlier in pregnancy were missed.²⁷ Congenital, genetic or valvular heart disease were rare causes of death. For those known to have these cardiac problems, inconsistencies in pre-conceptual advice and lack of integrated obstetric / cardiac specialist perinatal care, were highlighted in the 2019 MBRRACE-UK report.²⁷

3.5 Communicable disease

In the 2015–17 triennium, sepsis was the fifth most common cause of overall mortality. Death from sepsis includes genital tract and pregnancy related infections, influenza and pneumonia.²⁷ There was a non-statistically significant increase in deaths from sepsis compared to previous years in the 2016-18 MBRRACE report.⁴ Changes in the immune system, heart and lungs during pregnancy mean that pregnant women are more prone to severe illness from influenza viral infections and this risk is further increased in the third trimester.²⁸ Comparing the trienniums 2011–13 to 2016–18, influenza-related deaths have reduced from 0.38 to 0.09 per 100 000 maternities and this is attributed to universal influenza vaccination of pregnant women.^{4, 27}

The risks associated with COVID-19 infection in pregnancy have been highlighted during the current pandemic. SARS-CoV-2 is a new coronavirus but evidence to guide clinical care and identify risk factors is emerging. As with influenza, pregnancy-related changes in the immune response may cause severe COVID-19 symptoms, although the available evidence suggests pregnant women who develop coronavirus, are at no greater risk of becoming seriously unwell than other healthy adults, with the majority of pregnant women experiencing only mild or moderate symptoms.²⁹ However, avoidance of exposure to COVID-19, particularly in the third trimester is recommended. Such guidance will be much harder for women to adhere to where they live in overcrowded accommodation or are the sole carers of children. Furthermore, Black, Asian, and minority ethnic backgrounds and socioeconomic deprivation are independent risk factors for a higher risk of being infected and more severe COVID-19 infection.³⁰ These are also the same communities where rates of vaccine hesitancy in the UK have been reported as high, potentially increasing the risk of infection during pregnancy.³¹

The MBRRACE reports highlight the need to understand the differential burden of communicable (e.g. tuberculosis, HIV, hepatitis B) and non-communicable diseases (hypertension, heart disease, diabetes mellitus) by country of origin. Chronic heart disease related to rheumatic fever is rare in white British women, but more prevalent in low and middle income countries (LMIC), and in immigrants and older people in high income countries.³² In a recent European prospective cohort study of pregnancy outcomes of women with rheumatic heart disease (RHD), over 75% of women were originally from LMIC countries.³³ RHD is a very rare cause of death in the UK but the MBRRACE-UK reports cited above highlight the need for vigilance and recognition of cardiac symptoms in women born outside of the UK and particularly from Africa and Asia.

3.6 Body mass index

One of the key physical determinants for a poor maternal outcome is the mother's body mass index (BMI).³⁴ Maternal obesity and undernutrition are strongly linked to social deprivation, with increasing disadvantage in those with a BMI of 30 kg/m² and greater.³⁵⁻³⁷ Recent evidence has suggested that many women of reproductive age within the UK (and beyond) have suboptimal preconception nutrition^{38–39} and it has been recommended that any interventions to improve this should take a social determinants approach.⁴⁰

Obesity is associated with problems of early gestation, notably increased miscarriage and congenital abnormalities,⁴¹ increased risk of VTE during and after pregnancy, insulin resistance with overgrowth of the fetus, gestational diabetes and worsening of diabetic control for women with diabetes mellitus. In addition, obesity is linked to pregnancy-induced hypertension, pre-eclampsia and complications of

interventions such as caesarean section (wound infection, VTE).^{34,42} In the triennium 2003–05⁴³, 28% of maternal deaths were in women with a BMI greater than 30 kg/m², rising to more than one-third (34%) in the trienniums 2013–15⁴⁰ and 2015–17 (34%),²⁷ with a slight decrease in 2016-2018 to 29%.⁴ Data from 2015 reported that 20% of all pregnant women in the UK had a BMI over 30 kg/m²,⁴⁵ suggesting an increase in the mortality risk for obese pregnant women.

Links between social determinants, obesity and maternal outcomes have been reported. In a 3-year project undertaken by the Confidential Enquiry into Maternal and Child Health (CEMACH), published in 2010,³⁵ of those women living in England with a BMI more than 35 kg/m², 34% lived in the most deprived quintile of socioeconomic deprivation (by postcode) compared with 26.7% by all maternities. Black, Asian, and minority ethnic women were over-represented within this obese cohort, and more likely to have gestational diabetes and type II diabetes mellitus than white British women. This finding was supported in a large longitudinal study of English maternities. British Black and South Asian women, compared to White women, had a significantly higher incidence of obesity in the first trimester, with the highest odds observed in Pakistani women (OR 2.19, 95% CI 2.08 to 2.31, following adjustment for other population demographics).⁴⁶

Maternal micronutrient deficiency is associated with low birthweight, fetal growth restriction (FGR) and stillbirth. Although a causal relationship between this, social determinants and maternal mortality has not been established, low maternal calcium, vitamin D and zinc all appear to be associated with pregnancy disorders such as gestational hypertension, small-for-gestational-age fetuses and preterm birth. Women originally from settings where dietary micronutrient deficiency is more common than in the UK, may be particularly at risk of this form of undernutrition.⁴⁷

3.7 Mental health conditions

Since 2000, the overall rates of common mental disorders (CMDs) in England have steadily increased in women, compared to remaining mostly stable in men. Furthermore, CMDs were found to be more prevalent in Black women, those unemployed, and living in large households. Young women have also been identified as a particularly high-risk group for experiencing mental illness.⁴⁸

In the 2015–17 triennium, 'severe and multiple disadvantage' was highlighted in 6% of all deaths and included significant psychosocial challenges such as substance use, a mental health diagnosis and domestic abuse.²⁷ This rose to 8% in 2016-2018.⁴ Of the women who died in the 2016-2018 triennium, 20% were known to social services.⁴ Despite specific recommendations in previous MBRRACE-UK reports, under-reporting of a previous or pre-existing mental health problems remains a significant concern with information missing about mental health in 12% of women who died between 2016–2018.⁴ Missing data on key risk factors for mental illness were also noted in relation to substance use (6%) and domestic abuse (30%), although the latter does show an improvement from the 53% in 2015-2017.²⁷ While the role of physical and learning disabilities are outside the scope of this paper, earlier MBBRACE-UK reports have observed that a small number of women who died, also had a disability (particularly learning disability), compounding their highly vulnerable circumstances.⁴⁰ Consequently, cross-sector collaboration with national disability organisations (amongst others) is being undertaken in an effort to find ways to reduce mortality rates for these women with multiple disadvantage and complex mental health needs further.²⁷

Maternal suicide has been reclassified by WHO as a direct cause of maternal death. Since 2003, in consecutive UK maternal mortality reports, suicide rates have been static. Suicide is the leading direct cause of maternal death between 6 weeks and a year after the end of pregnancy (17%, 2016–2018).⁴ Causes of suicide are complex, but in the context of pregnancy, within the period 2014–2016 the women who died by suicide mainly lived in the most deprived areas, were multiparous, had a prior

history of mental illness and/or substance misuse, experienced domestic abuse, booked late and/or received inadequate antenatal care, were already known to social services, experienced a loss (e.g. through miscarriage, termination, and neonatal or infant death) and were white British/Irish citizens.²³

Although puerperal psychosis is rare and the association between puerperal psychosis and suicide has not been demonstrated in UK data, a systematic review identified three studies which reported 20 suicides in 220 puerperal psychosis patients.⁴⁹ These deaths often typically occur after 6 weeks post childbirth, reinforcing the importance that early recognition within primary and community health care services and access to specialist care is key to prevention of maternal and infant harm.

3.8. Smoking

Smoking during pregnancy is associated with maternal problems such as FGR, stillbirth, maternal hypertension and increased risk of venous thromboembolism.⁵⁰ Smoking during pregnancy is strongly linked to social deprivation and poor access to maternity services.⁵⁰⁻⁵¹ The 'Smoking Status at Time of Delivery' study for the period April 2017–March 2018, published in July 2018, showed that only 36 of 207 Clinical Commissioning Groups (CCGs) in England met the quality standard smoking rate of 6% at birth, with the UK average at 10.8%.⁵² Of significant concern was the minimal impact of health promotion and access to antenatal smoking cessation support in the most deprived areas of England, with a smoking rate of almost 26% at birth in NHS Blackpool, 22.1% in East Lincolnshire and 21.9% in NHS Durham Dales, Easington and Sedgefield. This was compared with 1.6% in NHS Central London (Westminster), and 2.9% in NHS Hammersmith and Fulham, NHS Wandsworth and NHS West London.

Smoking cessation is more prevalent among women with a higher level of education and income, where a strong link between access to early antenatal care and smoking cessation support is associated with successful cessation. Women from deprived backgrounds are less likely to engage early with smoking cessation support services during pregnancy and are therefore at greater risk of the adverse outcomes.⁵³

It is not known whether vaping, or e-cigarettes, during pregnancy is safe. Damaging effects on fetal growth and development have been demonstrated in animal models⁵⁴ and current UK smoking cessation guidance during pregnancy recommends the cessation of any smoking during pregnancy.⁵⁵ However, new evidence is emerging from adult smokers in general which suggests that e-cigarettes are likely to be significantly less harmful to a pregnant woman and her baby than continuing to smoke.⁵⁶ However, more research is being undertaken and is needed in this area.⁵⁷ Whether socially deprived women are more likely to vape during pregnancy is not known, however given that in the general UK population, vaping is more common in people experiencing higher levels of social deprivation,⁵⁸ potentially this pattern is also likely during pregnancy.

3.9. Alcohol misuse

The link between social determinants and substance misuse and how these factors contribute to adverse maternal outcomes is increasingly recognised, although some uncertainty still exists.

The latest MBBRACE-UK report did not differentiate between the types of substance use. However, 'Drug and alcohol/others' was a characteristic of 15% of the women who died in the 2016-2018 triennia (an increase from 12% in 2015–2017) between 6 weeks and 1 year after the end of pregnancy, in the UK.⁴

In a UK based study, over half of women in the first trimester reported alcohol intakes above the Department of Health (UK) guidelines of 2 units/week or less.⁵⁹ Furthermore, it has been found that

alcohol use during pregnancy is common⁶⁰: almost 16% of women resident in Europe consumed alcohol during pregnancy with large cross-country variations.⁶¹ Although most women would stop or reduce their levels of drinking when pregnancy was diagnosed, 12.5% continued to drink, even to binge levels.⁶¹

It is advised that women trying to conceive and those in the first three months of pregnancy should not drink alcohol.⁶² Whilst dependent/heavy alcohol use in pregnancy is uncommon, the evidence suggests that this excessive level of alcohol intake poses significant risks to the woman and the baby which include acute withdrawal syndromes, congenital and birth defects as well as longer term biopsychosocial harm.⁶³⁻⁶⁴

However, there is more uncertainty around the harm from lower levels of alcohol intake and the impact upon materno-fetal outcomes. Data from a recent meta-analysis demonstrated that there is a greater risk of miscarriage (odds ratio 1.19) in women exposed to alcohol during their pregnancy, compared to those that abstain. Furthermore, this risk was dose-mediated i.e., whereby each additional weekly drink was associated with a 6% increase risk of miscarriage (based on women consuming between 0-5 alcoholic drinks per week).⁶⁵ Another meta-analysis looking at low alcohol consumption and maternofoetal outcomes concluded that with the exception of birth size and gestational age, there was insufficient data to make any further robust conclusions.⁶⁴ Therefore, it has been argued that the position for mothers to completely abstain from alcohol during pregnancy is precautionary.⁶⁴

The links between social determinants and maternal alcohol use is unclear. Generally, though there is considerable variation in alcohol consumption by gender and socioeconomic status across Europe, with conflicting trends seen. For example, it has been consistently found that abstinence from alcohol is more likely in those with lower socio-economic status (SES). Other studies show that frequent consumption of alcohol, is more likely for those of higher SES, particularly for women. However, binge drinking and alcohol dependency are more likely in those with lower SES for both genders.⁶⁶

Alcohol misuse generally is strongly associated with domestic abuse; commonly associated with a breakdown in family and peer support, which are key to supporting positive maternity psychosocial outcomes; and may be associated with underlying or consequent mental health problems.⁶⁷ Women with heavy evident alcohol use would need to be cared for by joint obstetric and specialist substance misuse teams to supplement medical support for alcohol withdrawal treatments and psychosocial interventions to support abstinence, and if this isn't possible safer use.^{63, 66} Alcohol misuse may be a barrier to access to care for long term physical health conditions and is an independent risk factor for suicide.⁶⁸ National Institute of Health and Care Excellence (NICE) guidance⁶⁹ reinforces the need to address high population consumption through political and health promotion measures.

3.10 Illicit and prescribed drug misuse

An estimated 1% of pregnant women are problem drug users.⁷⁰ While the focus is usually on illicit use of heroin, crack cocaine, cannabis, amphetamines and a constantly changing array of 'street drugs', the growing phenomenon of dependence on prescribed opiates, especially in North America, is an example of 'hidden harm'.⁷⁰ The woman dependent on prescription synthetic opioid drugs (e.g. fentanyl) may not present as a stereotypical illicit drug user. Substance misuse among women who died in consecutive triennial reports is still implicated in maternal mortality in women with 'severe and multiple disadvantage'.⁴ Women who use illicit substances also commonly misuse alcohol compounding toxic effects and the risk of adverse maternal and fetal outcomes.⁷²

Pregnancy may provide an opportunity for positive change for women with substance use problems with motivations, including a healthier pregnancy, and improved physical and mental health generally postpartum. Initiatives include supported detoxification and abstinence or stabilisation with carefully monitored opiate agonist therapy (usually methadone). Access to social and housing advice and support can also be offered in joint specialist obstetric and substance use clinics staffed by multidisciplinary teams. Sudden withdrawal, especially in the third trimester, may be associated with fetal distress and stillbirth, and neonatal abstinence syndrome requires joint neonatal and specialist support.⁷³

3.11 Domestic abuse

The framework for thinking about the social determinants of maternal outcomes places mothers at the centre of their family, peers and social networks. It is these relationships that will often influence women's pregnancy-related behaviours and outcomes and impact on their ability to make informed and autonomous decisions, for example their freedom to make reproductive choices and ensure they have access to the services and care needed while pregnant and as a new mother.⁷⁴

Pregnancy and the postpartum period have been identified as risk factors for domestic abuse, and there have been increased reports of intimate partner violence during the pandemic.⁷⁵ The woman's partner is typically the most frequent perpetrator of domestic abuse. Between 2009 and 2013, a total of 36 mothers were murdered and 86% were murdered by a partner or a former partner.²¹ Pregnancy can be a period in a woman's life when the risk of domestic abuse is increased, in its frequency and/or intensity.²¹ Women who are the victims of abuse are known to suffer a number of poor outcomes, including preterm labour, antepartum haemorrhage, stillbirth, low birthweight babies, and other undetected obstetric complications as a consequence of not being able to attend/seek care.⁷⁶ The psychological consequences of the abuse can also be substantial. The 2007 Confidential Enquiry into for Maternal and Child Health, which covered the triennium 2003–05, reported that of the 33 mothers who died from suicide between 2003 and 2005, 42% were living with domestic abuse.⁴³

Most recently, of the 217 women who died from both indirect and direct causes, during or up to six weeks after pregnancy between the period 2016–2018, 19 (9%) had experienced domestic abuse either before and/or during pregnancy.⁴ But as mentioned earlier, the lack of recording in women's medical notes, resulted in a high level of missing data during this period (30%), although this was less than the levels of missing data found during the previous triennium which was 53%.²⁷ Changes in care delivery as a result of the pandemic is a current concern for this group of women, whereby the move to virtual or telephone appointments, and reduced healthcare interactions further inhibits the opportunity for information of domestic abuse to be identified and recorded in medical notes and for women to safely disclose, receive support, and have a means to ask for obstetric or midwifery help if control extends to access to phones.⁷⁷

The links between social inequality and domestic abuse have highlighted the vulnerable social circumstances and position of multiple disadvantage that some pregnant women find themselves in. For example, of the 42 women who had three or more indicators of social disadvantage in the 2016-2018 MBBRACE-UK report, 36 had experienced domestic abuse.⁴ In the UK, during the period 2009–13, the murder rate from domestic abuse was identified as almost two and a half times higher in Black, Asian, and minority ethnic women than in white women.²¹ Of the 23/36 mothers who were murdered during this same period between 6 weeks and 1 year after the end of pregnancy, the largest group (34.8%) lived in the most deprived areas, i.e. the fifth quintile measured using the Index of Multiple Deprivation (IMD). They were also younger (aged 20–24 years) and first-time mothers.²¹ The aforementioned CMACE 2007 document reported similar findings of the 19 women who died from

domestic abuse, whereby, 15 were known to local social services departments, nine had previous children in care, five did not speak English, three engaged in sex work and two had been in prison.⁴³

3.12 Female genital mutilation

Female genital mutilation (FGM) is prevalent in some of the countries of origin from which women are migrating or seeking asylum. In one report²¹ some of the mothers who died had undergone FGM, although this was not considered to be a contributory factor in their death, including those who were murdered.

FGM is not carried out for therapeutic reasons but is culturally and socially determined. Women with FGM are significantly more likely than those without to have adverse obstetric outcomes, and the risk of such outcomes increases if the woman has undergone more extensive FGM.⁷⁸ Fear about their immigration status, feelings of shame related to their experiences of sexual assault and exploitation and of the perinatal interventions associated with their FGM status are also potential barriers to seeking care.¹⁵ However, FGM may not be associated with an increased incidence of adverse obstetric and foetal morbidity or mortality when managed in high income countries such as UK.⁷⁹

4. Institutional and structural risk factors for adverse pregnancy outcomes and their links to social determinants

4.1 Socioeconomic position

There is a strong relationship between low socioeconomic position (SEP) and poor maternal outcomes. SEP is a term that denotes how social and economic factors influence the position an individual or group holds within society.⁸⁰ Other terms used to describe this include socioeconomic status and social class, and there are debates about their appropriateness.⁸¹ Typically, factors such as income, poverty, deprivation, wealth and education are indicators for SEP, although this may differ depending on the nature and purpose of the work, or the region and country in which SEP is being measured and analysed.⁸⁰

To date, the CEMACH and MBRRACE-UK reports have had to rely heavily on the Indices of Multiple Deprivation (IMD) as a measure of SEP in society because data are not available at a national level to provide denominator data on any other measure of socioeconomic status. When this is used, a stark social gradient is evident with the rate of maternal mortality highest among women living in the most deprived areas. For example, during the period 2016–18, of the 217 women who died during or up to 6 weeks after pregnancy from all causes, most lived in the most deprived area (37%) compared to the least deprived area (7%).⁴ The same trend was observed during the period 2015–17, whereby of the 209 women who died during or up to 6 weeks after pregnancy from all causes, 32% lived in the most deprived area compared, to 7% from the least deprived area.²⁷

In terms of occupation, a similar trend has been observed over the last three trienniums where during the period 2016-2018, 65% of women were living in a household where either themselves and/or their partner were employed⁴, compared to 60% in 2015–17²⁷, and 58% of women during 2013–15.⁷⁰ However, because the data were not available, the reports did not draw distinctions based on a woman's occupation only and there was no recording of women's levels of education. It was encouraging that the levels of missing data for occupation status had declined during the most recent triennium to 17%, from 20% during 2015-2017 and 32% during the period 2013–15.⁴⁴ However, the extent of missing data for the IMD of postcode of residence during these periods were similar at 16%, 17% and 18% respectively. It is reported that the adverse economic effects of the COVID-19 pandemic

have also disproportionately affected women, placing them at greater risk of unemployment, and reduced their access to sexual and reproductive health services, which may lead to them suffering longer term health impacts.⁸²

4.2 Partners and unemployment

Previous data have suggested that pregnant women appear to be at increased risk of dying if their partner is unemployed or in an unclassified occupation.¹⁵ For example, between 2003 and 2005 the maternal mortality rate for this group was 68.5 per 100 000 maternities, 7.4 times higher than the equivalent rate of 9.2 for all women with partners in employment. Women who were without a partner, regardless of social class, during this period were more than four times more likely to die than those with a partner in employment (maternal mortality rates of 38.6 versus 9.2 respectively).^{15, 43}

4.3 Education status

Education status often coincides with SEP. Even after adjusting for key confounding factors, women with lower levels of education appear at greater risk of severe maternal outcomes. This is particularly true for women from countries that have poorer markers of social and economic development.⁸³ One reason may be that women with lower levels of education are more likely to find themselves in adverse social circumstances which are closely connected, e.g. low-income occupations, abusive relationships and living in more deprived areas. In addition, they may be less likely to have the health literacy to engage with or access the information and care pathways that would help optimise pregnancy outcomes, such as reproductive health information and antenatal care. In LMIC countries there is a strong association between education status and the use of maternal services.⁸⁴ It is worth noting that with the exception of Census data, few data were collected in the UK about mothers' education status in pregnancy, although it is widely used in other countries. The disruptive effects of the COVID-19 pandemic on education can be expected to contribute to future disadvantage for some women.

4.4 Ethnicity

Ethnicity is very difficult to classify. How data is collected varies and so it is difficult to unpick relationships. One of the main reasons for this is because even within the existing classifications and definitions of ethnicity, there remains significant cultural differences and different terminologies used. This is because ethnicity is socially constructed and it is this societal classification of assigning individuals to one group or another which can negatively impact upon pregnant women at many different levels, for example at a personal, institutional, and structural level.⁸⁵⁻⁸⁶ Some of these impacts are described below but they are also described further under the sections on migration and asylum and health service factors.

From the information available, ethnicity appears to be a key social determinant of maternal mortality in many of the studies carried out globally. For example, non-Hispanic Black women in the USA die from pregnancy-related causes at a rate three to four times that of non-Hispanic white women.⁸⁷

In the UK, between 2016 and 2018, the risk of dying in pregnancy or up to 6 weeks postpartum was significantly higher among women from Black, Asian, and mixed ethnic backgrounds compared with white women.⁴ This risk was greatest for Black women being four times higher than white women (RR 4.35; 95% CI 2.77 to 6.62). Women of mixed ethnicity and women from Asian backgrounds (excluding women of Chinese and other Asian ethnicities) are also at higher risk than white women [(Mixed ethnicity RR 3.19, 95% CI 1.35.6.50) (Asian RR 1.86, 95% CI 1.19-2.830)]. There are no statistically

significant differences in the relative risk ratios between the recent triennia, suggesting that the inequality gap is neither improving nor worsening over time.⁴ The observation that women from Black, Asian, and minority ethnic backgrounds compared with white women have an increased risk of maternal mortality has been consistent across many reports and across other high-income countries too.^{4,88}

Reasons for these differences are unclear but research is being undertaken to explore these disparities.²⁷ An unmatched case-control study analysed data from women who experienced severe morbidity during the peripartum period (cases) and women who delivered immediately before the cases in the same hospital (controls) using the United Kingdom Obstetric Surveillance System (UKOSS), February 2005–January 2013. While an increased risk of severe maternal morbidity among women of ethnic minority backgrounds (Black African, Caribbean, Bangladeshi, other non-white and Pakistani women) compared with white European women was observed, this could not be explained by socioeconomic or smoking status or by BMI. Instead, the independent risk factors included anaemia in current pregnancy, having a previous pregnancy problem, experience inadequate provision of antenatal care services, pre-existing medical conditions, parity of more than three, and being younger or older.⁸⁹

4.5 Migration and asylum

The demographics of women who give birth in the UK are changing as are the risk factors for maternal mortality. Since 1969, the percentage of live births to non-UK born mothers has more than doubled over the last 50 years from 11.7% to 28.7% and it is now at the highest level since these records began.⁹⁰

This is important, because in the 2019 MBRRACE-UK report,²⁷ nearly a quarter of the women who died between 2015 and 2017 (from all causes) were born outside the UK (23%) and 42% of these women did not have UK citizenship. In the 2020 MBRRACE-UK report,⁴ the figures were similar at 26% and 36% respectively although women's origin of birth had no significant association on mortality rate across both of these trienniums. Most net migration to the UK has been from citizens living outside the European Union and this has been steadily increasing since 2013.⁹¹

The connection between a woman's place of birth and socioeconomic factors/co-morbidities is not clear, but women seeking asylum potentially have the highest unmet physical and psychological needs. They are usually socially disconnected from peer and family support, have significant language barriers, arrive having experienced significant trauma, are usually moving around the UK into 'no choice' accommodation, that provides substandard housing away from appropriate health services, have insecure/inadequate incomes, and most importantly do not know how and from where to access help.⁹² A study exploring the factors that may influence health-seeking behaviour among pregnant migrant women in England, found that poor information provision (including a lack of interpreters and culturally appropriate information), communication difficulties with healthcare staff, transport costs and lack of transport, dispersal/relocation issues, concerns about their immigration status, not seeing the service as useful or not being well enough to attend were all barriers to accessing antenatal care.⁹²⁻⁹³ The other relevant factor is that many migrant women may fear being charged for their maternity care, or being identified, and so do not present to services.^{27,94}

Some evidence also suggests that the impact of the acculturation process can result in migrant women adopting unhealthy behaviours prevalent in their new environment (particularly in relation to smoking and alcohol). These behaviours appear to worsen the longer they reside in the new country which can contribute to poorer maternal outcomes.⁹⁵

4.6 Environmental factors

The link between social determinants and deprivation cannot be discussed without considering environmental factors and their potential impact on adverse maternal outcomes, which are increasingly recognised.⁹⁶ Although causality has not been established, there is emerging evidence of maternal adverse effects of air and noise pollution on pre-eclampsia and gestational hypertension.⁹⁷ One report reviewed 16 meta-analyses of environmental pollutants that adversely affects pregnancy outcomes. They noted that although studies were of low methodological quality, the meta-analyses suggested statistically significant associations between: 1) environmental tobacco smoke (ETS) and stillbirth, birth weight and congenital anomalies, 2) maternal exposure to particulate matter with aerodynamic diameter between 2.5 and 10 µm and preterm birth, outdoor air pollution and some congenital anomalies, 3) indoor air pollution from solid fuel use and stillbirth and birth weight, 4) exposure to polychlorinated biphenyls and birth weight, 5) disinfection by-products in water and stillbirth, small-for-gestational age (SGA) and possibly some congenital anomalies, 6) occupational exposure to pesticides and solvents and some congenital anomalies, and agent orange and some congenital anomalies.⁹⁸

4.7 Health service factors

NICE recommends that ideally pregnant women should access antenatal care by 10^{+0} weeks of gestation.⁹⁹ Delayed access to antenatal care, referred to as 'late booking' in the UK, is linked to worse maternal outcomes. In addition to missing routine antenatal checks and tests of maternal and fetal wellbeing, women may not receive timely antenatal information (e.g. around lifestyle advice, folic acid supplementation and screening tests) and preventative interventions such as influenza vaccination. Between 2016 and 2018, of the women who died who received any antenatal care (n = 187); 61% of these did not receive the recommended level of care i.e. booked at 10 weeks or less, with no routine antenatal visits missed).⁴ This figure was similar to that observed in the triennium 2015–17 (63%).²⁷

There is a strong link between social determinants and delayed access to care.¹⁰⁰ Key characteristics identified for poor or non-attendance include higher parity and immigrant status.¹⁰¹ A review of experiences of disadvantaged women in the UK maternity system found that, amongst other factors, access was impacted by availability of interpreters, education, practical support and lack of continuity of care models.¹⁰² Other studies^{103–107} from in and outside the UK have found that common characteristics of 'late bookers' to antenatal care are ethnicity, young age, low income, low educational and support levels, asylum seekers and substance misuse – all affecting women with severe and multiple disadvantage.¹⁰⁷ Learning disability also appears to have played a role for a small number of women who had undertook infrequent and intermittent access to pregnancy care services.⁴⁴ The barriers to antenatal care attendance that women on low incomes, who are often from the poorest community's face has also led for 'poverty proofing' and identifying ways to remove some of the hidden costs associated with pregnancy.¹⁰⁸

A qualitative study¹⁰⁹ proposed a novel taxonomy to describe the reasons for late or non-booking of antenatal care. The key factors were: medical and system errors within the NHS (e.g. failure to send appointment letters), women simply 'not knowing they were pregnant (e.g. because of experiencing no traditional pregnancy symptoms)' or 'knowing they were pregnant but having decided to postpone attendance' (e.g. because of the perceived lack of value of antenatal care). Furthermore, a lack of reproductive health knowledge was a cross-cutting theme, which compounded other barriers to timely access to care. Some of these themes have been observed when exploring reasons for non-attendance of antenatal care services in LMIC too – particularly the failure for antenatal care services to be delivered in a way which recognises the socio-economic and cultural contexts in which pregnancy and birth are perceived and experienced.¹¹⁰

Access to postnatal care is also affected by the same social determinants. The Marmot review in 2010 highlighted the importance of integrated antenatal and postnatal policies, with access to high quality parenting programs, early childhood education and childcare to ensure every child is given the best start in life.¹ An updated look at these recommendations in 2020 by the same group found persistent inequalities in the UK in giving every child the best start, noting increased rates of child poverty since the original report.¹¹¹

Finally, access to antenatal and postnatal care cannot be considered without consideration of the environment within which pregnant women have to engage. Recent reports raise concerns about systematic problems within maternity services which may account for the racial and ethnic disparities observed.^{86, 112-117} For example because of institutional racism, and because mothers from ethnic minority groups are more likely to experience and report discrimination and bias in their care and report distrust of medical professionals, they are prevented from being able to access timely healthcare.¹¹⁸⁻¹¹⁹

4.8 Governance and law

According to the Human Rights Act 1998, the right to liberty and security, with freedom from torture, cruel, inhumane or degrading treatment is a fundamental human right.¹²⁰ Abuse of such rights can happen from gender-based violence, FGM, forced marriage, domestic abuse and sexual violence. This disproportionately affects women from minority ethnic groups and has direct links to poor pregnancy outcomes including maternal deaths.¹²¹ A report by Birthrights and Birth Companions in 2019 relating to women with severe and multiple disadvantage identified potential to breach these women's rights to safe and appropriate maternity care, respectful and dignified treatment, autonomy, choice and consent, respect for private and family life and equality, all of which relate to Articles 2 and 3 of the European Convention on Human Rights (right to life and prohibition on inhumane or degrading treatment), Article 8 (private and family life) and Article 14 (non-discrimination).¹⁰⁷

Global and UK government prioritisation of women's health has a major impact on maternal mortality. The Female Genital Mutilation Act of 2003, as amended by the serious crime act of 2015, has made this practice illegal in the UK. Anyone found guilty of an offence under this Act faces a maximum penalty of up to 14 years in jail. The widespread implementation of the law, coupled with aggressive advocacy, guidance development, and public enlightenment on a global scale have contributed to the sharp decline in this practice in many countries of the world.¹²²

Globally, unsafe abortion is estimated to account for 48 000 maternal deaths each year. There is evidence that maternal deaths dramatically reduced following the legalisation of abortion in the UK Abortion Act 1967, coupled with the provision of readily accessible and safe abortion services.¹²³ This is supported by global data¹²⁴ showing maternal death rates rise with restriction of access to safe abortion services, and conversely, death rates fall dramatically when safe abortion services are available.

5. Opinion

The complex causal pathways linking social inequalities and maternal mortality are extremely difficult to disentangle with the current levels of evidence available. Whilst members of the obstetric and midwifery multidisciplinary team may have limited power to modify social determinants of health, awareness of these impacts is important so that at-risk groups can be identified and have access to care as early as possible. A number of reports have been produced that provide recommendations to improve maternal outcomes, focussing on service delivery, policy changes and potential areas of research that particularly reflect the needs of women facing multiple disadvantage. These focus specifically on the UK context as the interventions/policy recommendations may be different in the global context.

• General care of pregnant women.

- Sharing of records between healthcare providers (e.g. between primary, secondary and social care), especially if they contain useful background information (e.g. history of selfharm, domestic abuse, children in care).^{21, 107}
- Extra vigilance by general practitioners, midwives, obstetricians and gynaecologists regarding recent migrants who may enter the country with pre-existing medical conditions that are prevalent in their country of origin, but rare in the UK.¹²⁵
- Policymakers should provide national guidance for the care of women with multiple physical and mental health morbidities and social factors prior to, during and after pregnancy.^{21, 107}
- Development of care pathways, that include out-reach work and centre around the provision of continuity of care, to better support access and engagement with antenatal and postnatal care for women with multiple disadvantage (e.g. for women whose infants are at risk of removal by social care¹⁰⁷ and/or where dispersal and relocation policies, or accessibility of virtual appointments may prevent/hinder access to timely antenatal and postnatal care).^{126,92}
- Development of care pathways that provide easy access to long-acting contraception to reduce unwanted pregnancies and pregnancy terminations.

• Improving the collection and reporting of maternal outcome data.

- Developing training packages to provide better support for midwives, obstetricians and gynaecologists to take a structured psychosocial history, beyond the physical and medical individual characteristics presented by the pregnant woman.
- Improving the recording of social determinants to help understand their contribution to overall maternal mortality.¹⁰⁷ It has recently been proposed that the distinction between direct and indirect causes is counterproductive, given that the goal of obstetric care should be on reducing all preventable deaths.¹²⁷ Taking a social determinant perspective would support this argument as health inequalities play an important role in maternal deaths in both causes. An additional problem lies in the classification of causes and the extent to which a death is traced back to an initial morbid event i.e. the causes of the causes.⁹⁴ Consequently, to better facilitate the identification of risk factors and achieve uniformity in data collection nationally and internationally, improving the way that maternal deaths are classified and recorded is warranted.

• Investment in preventative and intervention measures.

- Research investment by research councils/NIHR into preventative measures that can provide women with improved education/information about sexual reproductive health, preconception, antenatal, intrapartum and postnatal care.¹²⁸
- Investment by research councils/NIHR in establishing interventions that seek to improve the quality of, and access to, sexual reproductive health (e.g. contraception), preconception, antenatal, intrapartum and postnatal care²¹ (aimed at mothers, families and peer groups, including women with pre-existing medical problems and migrant women).¹²⁹ For example, improving access via new technology and working with hard to reach groups of pregnant women to deliver care in novel ways.¹²⁸
- Research investment to explore health care professional and health services role in enabling access to antenatal care for women from all backgrounds, for example cultural sensitivity and cultural congruent care (e.g. the role of unconscious/conscious bias).⁸⁶

Further to these recommendations, the following should be considered to improve future assessment, management and clinical outcomes of these women:

- A holistic multidisciplinary approach to the care of pregnant women by midwives, obstetricians and gynaecologists taking into account women's preconception and social circumstances including enhanced educational programmes at adolescent age to highlight risks to maternal and offspring health outcomes of adverse social determinants.⁹²
- Collecting detailed social determinant data for all pregnant women as part of routine data collection that do not require data linkage, i.e. it will be readily available for things such as the investigation of mortality and morbidity. Recent reports and infographics are based upon insufficient social determinant data and therefore the information presented is not adequate to illustrate the highly complex relationships between social determinants and maternal mortality.⁴
- Formal systematic reviews are needed in this area, particularly in the UK context. Arising from our work on this SIP, the team are currently undertaking such a review to synthesise the literature in this area further.
- Developing new tools (e.g. questionnaires and screening systems) that are efficient and simple to use (i.e. allowing integration into electronic patient records) to facilitate the collection of more detailed data on social determinants as part of routine clinical practice and to inform MBBRACE-UK reports.
- Agreeing appropriate terminology and definitions for the collection of certain types of social determinant data (e.g. maternal death from domestic abuse, and disability). Current information systems cannot handle these types of data accurately.¹²⁹⁻¹³⁰ It is important to note that whilst we have used the word 'disability' this is based on how it has been classified in previous reports. Disability rights advocates and campaigners are urging a move towards the social model of disability, i.e. a person may have 'impairments', such as mobility issues or a chronic mental health condition but it is not these which disable them; rather it is societal attitudes or physical obstacles which act as barriers to participation/inclusion/better outcomes. Whilst using such terminology to record and discuss disability has been agreed and utilised by the Welsh Government in its strategy/policy documents, adhering to this social model of disability more widely is particularly recommended.¹³¹

In summary, adverse social determinants of health affect maternal clinical outcomes, leading to an increased risk of maternal mortality and morbidity. Although the numbers of pregnant women who die in the UK are low, identifying and mitigating the multiple factors that contribute to the web of disadvantage will improve birth outcomes and reduce maternal morbidity as well as mortality. In doing so, however, it is important that disadvantaged women are not stigmatised and subjected to excessive and unwarranted interventions. Identifying and chronicling women's psycho-social history is essential to drive improvements in health outcomes, i.e. taking a life course approach to the assessment of the pregnant woman. The recent experiences during the coronavirus pandemic that highlights emerging links to increased adverse outcomes in Black, Asian, and minority ethnic groups and those with an increased BMI illustrate that if we are to improve our understanding and care for diseases and conditions related to pregnancy, we will need to develop interventions that address adverse social determinants of health as well as manage the conditions themselves in order to improve maternal outcomes for all pregnant women – especially those at most disadvantage.

References

- 1. Strategic Review of Health Inequalities in England post-2010. (2010). Fair society, healthier lives: The Marmot review. Retrieved from http://www.instituteofhealthequity.org/resources-reports/fair-society-healthy-lives-the-marmot-review/fair-society-healthy-lives-the-marmot-review-full-report.pdf
- 2. Marmot M. Social determinants of health inequalities. Lancet 2005;365:1099–104.
- 3. Loudon I Death in Childbirth: An International Study of Maternal Care and Maternal Mortality 1800-1950
- 4. Knight M, Bunch K, Tuffnell D, Shakespeare J, Kotnis R, Kenyon S, Kurinczuk JJ (Eds.) on behalf of MBRRACE-UK. Saving Lives, Improving Mothers' Care Lessons learned to inform maternity care from the UK and Ireland Confidential Enquiries into Maternal Deaths and Morbidity 2016-18. Oxford: National Perinatal Epidemiology Unit, University of Oxford 2020.
- 5. Kayem G, Kurinczuk J, Lewis G, Golightly S, Brocklehurst P. Risk factors for progression from severe maternal morbidity to death: a national cohort study. PLoS One 2011;6:e29077.
- 6. Nair M, Kurinczuk JJ, Brocklehurst P, Sellers S, Lewis G, Knight M. Factors associated with maternal death from direct pregnancy complications: a UK national case–control study. BJOG 2015;122:653–62.
- 7. Nair M, Nelson-Piercy C, Knight M. Indirect maternal deaths: UK and global perspectives. Obstet Med 2017;10:10–15.
- 8. Nair M, Knight M, Kurinczuk JJ. Risk factors and newborn outcomes associated with maternal deaths in the UK from 2009 to 2013: a national case–control study. BJOG 2016;123:1654–62.
- World Health Organization. The WHO application of ICD-10 to deaths during pregnancy, childbirth and the puerperium: ICD MM. Geneva: WHO; 2012 [https://www.who.int/reproductivehealth/publications/monitoring/9789241548458/en/]. Accessed 14th May 2019.
- Dahlgren G, Whitehead M; Institute for Futures Studies. Policies and strategies to promote social equity in health. Background document to WHO – Strategy paper for Europe. Arbetsrapport 1991;2007:14.
- 11. World Health Organization. Closing the gap: policy into practice on social determinants of health. Geneva: WHO; 2011 [https://apps.who.int/iris/handle/10665/44731]. Accessed 14th May 2020.
- The United Nations Development Programme. Discussion Paper. A Social Determinants Approach to Maternal Health. 2011. https://www.undp.org/content/dam/undp/library/Democratic%20Governance/Discussion%2 0Paper%20MaternalHealth.pdf Accessed 30th July 2021.
- Office for National Statistics. Birth characteristics in England and Wales: 2018 [https://www.ons.gov.uk/peoplepopulationandcommunity/birthsdeathsandmarriages/livebir ths/bulletins/birthcharacteristicsinenglandandwales/latest]. Accessed 13th May 2020.
- 14. Office for National Statistics. Conceptions in England and Wales: 2018 [https://www.ons.gov.uk/peoplepopulationandcommunity/birthsdeathsandmarriages/conce ptionandfertilityrates/bulletins/conceptionstatistics/2018]. Accessed 13th May 2020.
- Fisher S. Social inequalities in maternal and perinatal mortality: a summary of research. NCT New Digest 2008;44:18–26 [https://www.nct.org.uk/sites/default/files/related_documents/Social_inequalities_0.pdf]. Accessed 13th May 2020.
- 16. Mills M, Rindfuss RR, McDonald P, te Velde E. Why do people postpone parenthood? Reasons and social policy incentives. Hum Reprod Update 2011;17:848–60.
- 17. McCall SJ, Nair M, Knight M. Factors associated with maternal mortality at advanced maternal age: a population-based case-control study. BJOG 2017; 124(8): 1225-33.
- 18. Bradshaw J. Teenage births. York: Joseph Rowntree Foundation; 2006

[https://www.jrf.org.uk/sites/default/files/jrf/migrated/files/9781859355046.pdf]. Accessed 14th May 2020.

- Higginbottom GM, Mathers N, Marsh P, Kirkham M, Owen JM, Serrant-Green L. Young people of minority ethnic origin in England and early parenthood: views from young parents and service providers. Soc Sci Med. 2006 Aug;63(4):858-70. doi: 10.1016/j.socscimed.2006.03.011. Epub 2006 May 5. PMID: 16678322.
- Office for National Statistics. Pregnancy and ethnic factors influencing births and infant mortality: 2013. [https://www.ons.gov.uk/peoplepopulationandcommunity/healthandsocialcare/causesofdea
- th/bulletins/pregnancyandethnicfactorsinfluencingbirthsandinfantmortality/2015-10-14].
 21. Knight MNM, Tuffnell D, Kenyon S, Shakespeare J, Gray R, Kurinczuk JJ, (Eds). Saving lives, improving mothers' care surveillance of maternal deaths in the UK 2011–13 and lessons learned to inform maternity care from the UK and Ireland confidential enquiries into maternal deaths and morbidity 2009–13. Oxford: National Perinatal Epidemiology Unit, University of Oxford; 2015.
- 22. Knight M, Bunch K, Tuffnell D, Jayakody H, Shakespeare J, Kotnis R, Kenyon S, Kurinczuk JJ (Eds.) on behalf of MBRRACE-UK. Saving Lives, Improving Mothers' Care Lessons learned to inform maternity care from the UK and Ireland Confidential Enquiries into Maternal Deaths and Morbidity 2014-16. Oxford: National Perinatal Epidemiology Unit, University of Oxford 2018.
- 23. Mgaya, A.H., Massawe, S.N., Kidanto, H.L. et al. Grand multiparity: is it still a risk in pregnancy?. BMC Pregnancy Childbirth 13, 241 (2013). https://doi.org/10.1186/1471-2393-13-241)
- Sigle, Wendy (2016) Fertility and population change in the United Kingdom. In: Low Fertility, Institutions, and their Policies. Springer International Publishing, Cham, Switzerland, pp. 77-98. ISBN 9783319329970. Also available here: http://eprints.lse.ac.uk/86426/1/Sigle Fertility%20and%20population 2018.pdf
- Pezzulo, Carla et al. Geographical distribution of fertility rates in 70 low-income, lowermiddle-income, and upper-middle-income countries, 2010–16: a subnational analysis of cross-sectional surveys. The Lancet Global Health, Volume 9, Issue 6, e802 - e812
- 26. Dixon-Woods M, Kirk D, Agarwal S, Annandale E, Arthur T, Harvey J, et al. Vulnerable groups and access to health care: a critical interpretive review. Report for the National Co-ordinating Centre for NHS Service Delivery and Organisation R&D (NCCSDO). NCCSDO; 2005 [http://www.netscc.ac.uk/hsdr/files/project/SDO_FR_08-1210-025_V01.pdf]. Accessed 14th May 2020.
- 27. Knight M, Bunch K, Tuffnell D, Shakespeare J, Kotnis R, Kenyon S, Kurinczuk JJ (Eds.) on behalf of MBRRACE-UK. Saving Lives, Improving Mothers' Care - Lessons learned to inform maternity care from the UK and Ireland Confidential Enquiries into Maternal Deaths and Morbidity 2015-17. Oxford: National Perinatal Epidemiology Unit, University of Oxford 2019: p5-19.
- 28. Mak TK, Mangtani P, Leese J, Watson JM, Pfeifer D. Influenza vaccination in pregnancy: current evidence and selected national policies. The Lancet infectious diseases. 2008 Jan 1;8(1):44-52.
- 29. https://www.rcog.org.uk/en/guidelines-research-services/guidelines/coronaviruspregnancy/2
- 30. Coronavirus (COVID-19) infection and pregnancy Version 8: updated Friday 17 April 2020. Guidance for healthcare professionals on coronavirus (COVID-19) infection in pregnancy, published by the RCOG, Royal College of Midwives, Royal College of Paediatrics and Child Health, Public Health England and Public Health Scotland. https://www.rcog.org.uk/coronavirus-pregnancy
- 31. Razai MS, Osama T, McKechnie DGJ, Majeed A. Covid-19 vaccine hesitancy among ethnic minority groups. BMJ 2021; 372: n513.

- 32. Watkins DA, Johnson CO, Colquhoun SM, Karthikeyan G, Beaton A, Bukhman G, et al. Global, Regional, and National Burden of Rheumatic Heart Disease, 1990–2015. N Engl J Med 2017;377:713–22.
- van Hagen IM, Thorne SA, Taha N, Youssef G, Elnagar A, Gabriel H, ElRakshy Y, Iung B, Johnson MR, Hall R, Roos-Hesselink JW. Pregnancy outcomes in women with rheumatic mitral valve disease: results from the registry of pregnancy and cardiac disease. Circulation. 2018 Feb 20;137(8):806-16.
- 34. Scott-Pillai R, Spence D, Cardwell CR, Hunter A, Holmes VA. The impact of body mass index on maternal and neonatal outcomes: a retrospective study in a UK obstetric population, 2004–2011. BJOG 2013;120:932–9.
- 35. Centre for Maternal and Child Enquires (CMACE). Maternal obesity in the UK: findings from a national project. London: CMACE; 2010.
- 36. Walker IV, Cresswell JA. Multiple deprivation and other risk factors for maternal obesity in Portsmouth, UK. J Public Health (Oxf) 2019;41:278–86.
- 37. Ahmed T, Hossain M, Sanin KI. Global burden of maternal and child undernutrition and micronutrient deficiencies. Ann Nutr Metab 2012;61 Suppl 1;8–17.
- 38. Stephenson J, Heslehurst N, Hall J, Schoenaker DAJM, Hutchinson J, Cade JE, et al. Before the beginning: nutrition and lifestyle in the preconception period and its importance for future health. Lancet 2018;391:1830–41.
- 39. Fleming TP, Watkins AJ, Velazquez MA, Mathers JC, Prentice AM, Stephenson J, et al. Origins of lifetime health around the time of conception: causes and consequences. Lancet 2018;391:1842–52.
- 40. Barker M, Dombrowski SU, Colbourn T, Fall CHD, Kriznik NM, Lawrence WT, et al. Intervention strategies to improve nutrition and health behaviours before conception. Lancet 2018;391:1853–64.
- 41. Stothard KJ, Tennant PWG, Bell R, Rankin J. Maternal overweight and obesity and the risk of congenital anomalies: a systematic review and meta-analysis. JAMA 2009;301:636–50.
- 42. Wloch C, Wilson J, Lamagni T, Harrington P, Charlett A, Sheridan E. Risk factors for surgical site infection following caesarean section in England: results from a multicentre cohort study. BJOG 2012;119:1324–33.
- 43. Lewis G (ed). The Confidential Enquiry into Maternal and Child Health (CEMACH). Saving Mothers' Lives: reviewing maternal deaths to make motherhood safer 20032005. The Seventh Report on Confidential Enquiries into Maternal Deaths in the United Kingdom. London: CEMACH; 2007.
- 44. Knight M, Nair M, Tuffnell D, Shakespeare J, Kenyon S, Kurinczuk JJ (Eds.) on behalf of MBRRACE-UK. Saving Lives, Improving Mothers' Care. Lessons learned to inform maternity care from the UK and Ireland Confidential Enquiries into Maternal Deaths and Morbidity 2013–15. Oxford: National Perinatal Epidemiology Unit, University of Oxford; 2017
- 45. Health and Social Care Information Centre. Maternity Services Monthly Statistics July 2016, Experimental Statistics [http://content.digital.nhs.uk/pubs/msmsjul16exp]. Accessed 14th May 2020.
- 46. Heslehurst N, Sattar N, Rajasingham D, Rankin J, Batterham A, Wilkinson JR, Summerbell CD. Maternal obesity and ethnic groups: trends in 552 303 births over 13 years in England, UK. Archives of Disease in Childhood-Fetal and Neonatal Edition. 2010 Jun 1;95(Suppl 1):Fa33-4.
- 47. Black RE, Victora CG, Walker SP, Bhutta ZA, Christian P, de Onis M, et al.; Maternal and Child Nutrition Study Group. Maternal and child undernutrition and overweight in low-income and middle-income countries. Lancet 2013;382:427–51.
- 48. McManus S, Bebbington P, Jenkins R, Brugha T. (eds.) (2016) Mental health and wellbeing in England: Adult Psychiatric Morbidity Survey 2014. Leeds: NHS Digital.
- 49. Gilden J, Kamperman AM, Munk-Olsen T, Hoogendijk WJG, Kushner SA, Bergink V. Long-Term Outcomes of Postpartum Psychosis: A Systematic Review and Meta-Analysis. J Clin Psychiatry.

2020 Mar 10;81(2). pii: 19r12906. doi: 10.4088/JCP.19r12906. PubMed PMID: 32160423.

- 50. Dias-Damé JL, Cesar JA. Disparities in prevalence of smoking and smoking cessation during pregnancy: a population-based study. Biomed Res Int 2015;2015:345430.
- 51. Silveira MF, Matijasevich A, Menezes AM, Horta BL, Santos IS, Barros AJ, et al. Secular trends in smoking during pregnancy according to income and ethnic group: four population-based perinatal surveys in a Brazilian city. BMJ Open 2016;6:e010127.
- 52. Health and Social Care Information Centre. Statistics on Women's Smoking Status at Time of Delivery: England Quarter 1, April 2018 to June 2018 [https://digital.nhs.uk/data-and-information/publications/statistical/statistics-on-women-s-smoking-status-at-time-of-delivery-england/statistics-on-womens-smoking-status-at-time-of-delivery-england-quarter-1-april-2018-to-june-2018]. Accessed 14th May 2020.
- 53. O'Gorman C. Smoking in pregnancy: comparisons between data from the Infant Feeding Survey 2000 and 2005. Evid Based Midwifery 2010;8:136–42.
- 54. Orzabal MR, Lunde-Young ER, Ramirez JI, et al. Chronic exposure to e-cig aerosols during early development causes vascular dysfunction and offspring growth deficits. Transl Res. 2019;207:70-82. doi:10.1016/j.trsl.2019.01.001
- 55. RCOG document patient-information-leaflets%2Fpregnancy%2Fpi-smoking-in pregnancy.pdf&clen=502631&chunk=true
- 56. https://smokefreeaction.org.uk/wp-content/uploads/2019/08/2019-Challenge-Group-ecigsbriefing-FINAL.pdf
- 57. https://doi.org/10.1186/ISRCTN62025374
- 58. Green MJ, Gray L, Sweeting H, Benzeval M. Socioeconomic patterning of vaping by smoking status among UK adults and youth. BMC Public Health. 2020;20(1):183. Published 2020 Feb 10. doi:10.1186/s12889-020-8270-3
- 59. Nykjaer C, Alwan NA, Greenwood DC, et al Maternal alcohol intake prior to and during pregnancy and risk of adverse birth outcomes: evidence from a British cohort. J Epidemiol Community Health 2014;68:542-549
- 60. Popova S, Lange S, Probst C, Gmel G, Rehm J. Estimation of national, regional, and global prevalence of alcohol use during pregnancy and fetal alcohol syndrome: a systematic review and meta-analysis. Lancet Glob Health 2017;5:e290–e299.
- 61. Mardby Consumption of alcohol during pregnancy-A multinational European study Ann-Charlotte Mårdby 1, Angela Lupattelli 2, Gunnel Hensing 3, Hedvig Nordeng 4 Affiliations expand PMID: 28111037 DOI: 10.1016/j.wombi.2017.01.003)
- 62. https://www.rcog.org.uk/en/news/rcog-release-alcohol-in-pregnancy-discussed-in-updated-patient-information/
- 63. British Medical Association. Alcohol and pregnancy. preventing and managing fetal alcohol spectrum disorders. BMA; 2007 (updated 2016).
- 64. Mamluk L, Edwards HB, Savović J, Leach V, Jones T, Moore THM, Ijaz S, Lewis SJ, Donovan JL, Lawlor D, Smith GD, Fraser A, Zuccolo L. Low alcohol consumption and pregnancy and childhood outcomes: time to change guidelines indicating apparently 'safe' levels of alcohol during pregnancy? A systematic review and meta-analyses. BMJ Open. 2017 Aug 3;7(7):e015410. doi: 10.1136/bmjopen-2016-015410. PMID: 28775124; PMCID: PMC5642770.
- 65. Sundermann, A.C., Zhao, S., Young, C.L., Lam, L., Jones, S.H., Velez Edwards, D.R. and Hartmann, K.E. (2019), Alcohol Use in Pregnancy and Miscarriage: A Systematic Review and Meta-Analysis. Alcohol Clin Exp Re, 43: 1606-1616. https://doi.org/10.1111/acer.14124
- 66. Anderson P & Baumberg B (2006) Alcohol in Europe. London: Institute of Alcohol Studies. https://ec.europa.eu/health/ph_determinants/life_style/alcohol/documents/alcohol_europe .pdf)

67. WHO Publication 2006:

https://www.who.int/violence_injury_prevention/violence/world_report/factsheets/fs_intim ate.pdf

- 68. Brady J. The association between alcohol misuse and suicidal behaviour. Alcohol and Alcoholism. 2006;41:473–478.
- 69. National Institute for Health Care Excellence. Antenatal care for uncomplicated pregnancies. NICE clinical guideline 62. NICE; 2008 [Updated 2019].
- Hidden Harm Responding to the needs of children of problem drug users. The report of an inquiry by the advisory council on the misuse of drugs. 2003. https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/120620/hi dden-harm-full.pdf.
- 72. Forray A, Foster D. Substance Use in the Perinatal Period. Curr Psychiatry Rep 2015;17:91.
- 73. Keegan J, Parva M, Finnegan M, Gerson A, Belden M. Addiction in pregnancy. J Addict Dis 2010;29:175–191.
- 74. Oxaal Z & Baden S (1996) Challenges to women's reproductive health: maternal mortality. Report number 38. Bridge (Development-Gender).
- 75. Violence against women during covid-19 pandemic restrictions. BMJ 2020; 369 doi: https://doi.org/10.1136/bmj.m1712 (Published 07 May 2020)
- 76. Donovan, BM, Spracklen, CN, Schweizer, ML, Ryckman, KK, Saftlas, AF. Intimate partner violence during pregnancy and the risk for adverse infant outcomes: a systematic review and meta-analysis. BJOG 2016; 123: 1289– 1299.
- 77. www.rcm.org.uk%2Fmedia%2F4067%2Fidentifying-caring-for-and-supporting-women-at-riskof_victims-of-domestic-abuse-during-covid-19v1_13052020final.pdf&clen=191359&chunk=true
- 78. WHO Study group on female genital mutilation and obstetric outcome, Banks E, Meirik O, Farley T, Akanda O, Bathija H, et al.Female genital mutilation and obstetric outcome: WHO collaborative prospective study in six African countries. Lancet 2006;367:1835–1841.
- Balachandran AA, Duvalla S, Sultan AH, Thakar R. Are obstetric outcomes affected by female genital mutilation? Int Urogynecol J. 2018 Mar;29(3):339-344. doi:10.1007/s00192-017-3466-5. Epub 2017 Sep 9. PMID: 28889193
- 80. Galobardes B, Shaw M, Lawlor DA, Lynch JW, Smith GD. Indicators of socioeconomic position (part 1). J Epidemiol Community Health 2006;60:7–12.
- 81. Krieger N, Williams DR, Moss NE. Measuring social class in US public health research: concepts, methodologies, and guidelines. Annu Rev Public Health 1997;18:341–378
- Rose Cook & Damian Grimshaw (2021) A gendered lens on COVID-19 employment and social policies in Europe, European Societies, 23:sup1, S215-S227, DOI: 10.1080/14616696.2020.1822538
- 83. Tunçalp Ö, Hindin MJ, Adu-Bonsaffoh K, Adanu RM. Assessment of maternal near-miss and quality of care in a hospital-based study in Accra, Ghana. Int J Gynecol Obstet 2013;123:58–63.
- 84. Kalule-sabiti I, Amoateng AY, Ngake M. The Effect of Socio-demographic Factors on the Utilization of Maternal Health Care Services in Uganda. African Population Studies 2014;28:515525.
- 85. Ford, C. L., & Harawa, N. T. (2010). A new conceptualization of ethnicity for social epidemiologic and health equity research. Social science & medicine (1982), 71(2), 251–258. https://doi.org/10.1016/j.socscimed.2010.04.008
- 86. Position Statement: Racial disparities in women's healthcare, Royal College of Obstetricians and Gynaecologists.
 2020.www.rcog.org.uk%2Fglobalassets%2Fdocuments%2Fnews%2Fpositionstatements%2Fra cial-disparities-womens-healthcare-march 2020.pdf&clen=620250&chunk=true
- 87. Creanga AA, Syverson C, Seed K, Callaghan WM. Pregnancy-Related Mortality in the United

States, 2011-2013. Obstet Gynecol 2017;130:366-373.

- 88. Van den Akker T et al. Maternal mortality and morbidity in a migration perspective. Best Pract Res Clin Obs Gyn 2016.
- 89. Nair M, Kurinczuk JJ, Knight M. Ethnic Variations in Severe Maternal Morbidity in the UK A Case Control Study. PLoS One 2014;9:e95086.
- 90. https://www.ons.gov.uk/peoplepopulationandcommunity/birthsdeathsandmarriages/ livebirths/bulletins/parentscountryofbirthenglandandwales/2019
- 91. Office for National Statistics. Migration Statistics Quarterly Report: August 2020. ONS; 2020.
- 92. Feldman R. When maternity doesn't matter. Dispersing pregnant women seeking asylum. Maternity Action.
- 93. Newall D, Phillimore J, Sharpe H. Migration and maternity in the age of superdiversity. Pract Midwife 2012;15:20–22.
- 94. Maternity Action. What Price Safe Motherhood? Charging for NHS Maternity Care in England and its Impact on Migrant Women. Maternity Action; 2018.
- 95. Perreira K. M. (2008). Migration and health behaviour during pregnancy. BMJ (Clinical research ed.), 336(7652), 1027–1028. https://doi.org/10.1136/bmj.39540.596782.BE
- 96. Gray, S.C., Edwards, S.E., Schultz, B.D. et al. Assessing the impact of race, social factors and air pollution on birth outcomes: a population-based study. Environ Health 13, 4 (2014). https://doi.org/10.1186/1476-069X-13-4
- 97. Assibey-Mensah V, Glantz JC, Hopke PK, Jusko TA, Thevenet-Morrison K, Chalupa D, Rich DQ. Wintertime Wood Smoke, Traffic Particle Pollution, and Preeclampsia. Hypertension. 2020 Mar;75(3):851-858. doi: 10.1161/HYPERTENSIONAHA.119.13139.
- 98. Nieuwenhuijsen, M.J., Dadvand, P., Grellier, J. et al. Environmental risk factors of pregnancy outcomes: a summary of recent meta-analyses of epidemiological studies. Environ Health 12, 6 (2013). https://doi.org/10.1186/1476-069X-12-6).
- 99. National Institute for Health Care Excellence. Antenatal care for uncomplicated pregnancies. NICE clinical guideline 62. NICE; 2008 [Updated 2019].
- 100. Kapaya H, Mercer E, Boffey F, Jones G, Mitchell C, Anumba D. Deprivation and poor psychosocial support are key determinants of late antenatal presentation and poor fetal outcomes-a combined retrospective and prospective study. BMC Pregnancy Childbirth 2015;15:1–9.
- 101. Simpson H, Walker G. When do pregnant women attend for antenatal care? Br Med J 1980;281:104–107.
- 102. Rayment-Jones H, Harris J, Harden A, Khan Z, Sandall J. How do women with social risk factors experience United Kingdom maternity care? A realist synthesis. Birth 2019; 46(3): 461-74.
- 103. Rowe R & Garcia J. (2003). Social class, ethnicity and attendance for antenatal care: a systematic review. Journal of Public Health Medicine 25:113-119.
- 104. Alderliesten ME, Vrijkotte TGM, van der Wal MF, Bonsel GJ. Late start of antenatal care among ethnic minorities in a large cohort of pregnant women. BJOG. 2007;114:1232–1239. doi: 10.1111/j.1471-0528.2007.01438.x.
- 105. Raatikainen K, Heiskanen N, Heinonen S. Under-attending free antenatal care is associated with adverse pregnancy outcomes. BMC Public Health. 2007;7:268. doi: 10.1186/1471-2458-7-268.
- 106. Kupek E, Petrou S, Vause S, Maresh M. Clinical, provider and sociodemographic predictors of late initiation of antenatal care in England and Wales. BJOG. 2002;109:265–273. doi: 10.1111/j.14710528.2002.00524.x.
- 107. Birthrights and Birth Companions (2019). Holding it all together: Understanding how far the human rights of woman facing disadvantage are respected during pregnancy, birth and postnatal care. London [online].Accessed 25 April 2020 at https://www.birthrights.org.uk/wp-content/uploads/2019/09/Holding-it-all-together-Fullreport-FINAL-Action-Plan.pdf

- 108. McHardy F. For Mothers and Babies: Pregnancy and Antenatal Care Low income women's experiences during the 80's 90's and early 2000's. 2018. Poverty Alliance. Scotland.
- 109. Haddrill R, Jones GL, Mitchell CA, Anumba DOC. Understanding delayed access to antenatal care: a qualitative interview study. BMC Pregnancy Childbirth 2014;14:207.
- 110. Finlayson K, Downe S (2013) Why Do Women Not Use Antenatal Services in Low- and Middle-Income Countries? A Meta-Synthesis of Qualitative Studies. PLOS Medicine 10(1): e1001373. https://doi.org/10.1371/journal.pmed.1001373
- 111. Health Equity in England: The Marmot Review 10 Years On (2020). Institute of Health Equity. https://www.health.org.uk/publications/reports/the-marmot-review-10-years-on
- 112. J. Dovidio et al., Racial biases in medicine and healthcare disparities (2016) and Center for American Progress, Eliminating Racial Disparities in Maternal and Infant Mortality: A Comprehensive Policy Blueprint (2019)
- 113. K Hoffman et al., Racial bias in pain assessment and treatment recommendations, and false beliefs about biological differences between blacks and whites (2016)
- 114. J. Aronson et al., Unhealthy Interactions: The Role of Stereotype Threat in Health Disparities (2013)
- 115. B. Byrne et al., Ethnicity, Race and Inequality in the UK: State of the nation (2020)
- 116. https://maternityaction.org.uk/midwivesreport/
- 117. Briefing on Black maternal healthcare and mortality. Westminster Hall debate, 6.15pm 19 April 2021
- 118. Henderson, J., Gao, H. & Redshaw, M. Experiencing maternity care: the care received and perceptions of women from different ethnic groups. BMC Pregnancy Childbirth 13, 196 (2013). https://doi.org/10.1186/1471-2393-13-196
- 119. Vedam, S., Stoll, K., Taiwo, T. K., Rubashkin, N., Cheyney, M., Strauss, N., . . . & the GVtM-US Steering Council. (2019). "The Giving Voice to Mothers study: Inequity and mistreatment during pregnancy and childbirth in the United States". Reproductive Health, June 11, 1-18.DOI: 10.1186/s12978-019-0729-2
- 120. The Human Rights Act 1998. Available from: http://www.legislation.gov.uk/ukpga/1998/42/contents.
- 121. Maternity Action. Mothers' voices. London: Maternity Action;2018.
- 122. Koski A, Heymann J. Thirty-year trends in the prevalence and severity of female genital mutilation : a comparison of 22 countries. BMJ Glob Health 2017;2:e000467.
- 123. Faúndes A, Shah IH. Evidence supporting broader access to safe legal abortion. Int J Gynecol Obstet 2015;131:S56–S59.
- 124. Grimes DA, Benson J, Singh S, Romero M, Ganatra B, Okonofua FE, et al.Unsafe abortion: the preventable pandemic. Lancet 2006;368:1908–1919.
- 125. Public Health England. Assessing new patients from overseas : migrant health guide. London: PHE; 2019.
- 126. https://www.birthrights.org.uk/wp-content/uploads/2020/05/Birthrights-Covid-19-Maternity-Briefing-women-and-equalities-committee-final.pdf
- 127. van den Akker T, Nair M, Goedhart M, Schutte J, Schaap T, Knight M, et al. Maternal mortality: direct or indirect has become irrelevant. Lancet Glob Heal 2017;5:e1181–e1182.
- 128. https://www.rcog.org.uk/better-for-women
- 129. Bambra C, Gibson M, Sowden A, Wright K, Whitehead M, Petticrew M. Tackling the wider social determinants of health and health inequalities: Evidence from systematic reviews. J Epidemiol Community Health 2010;64;284–291.
- 130. Espinoza H, Camacho AV. Maternal death due to domestic violence : an unrecognized critical component of maternal mortality. Rev Panam Salud Publica 2005;17:123–129.84.National Institute for Health and Care Excellence. Domestic violence and abuse: multi-agency working. NICE Public Health Guideline [PH50]. London: NICE; 2014.
- 131. Welsh Government Consultation Document Action on Disability: the Right to Independent

Living (2018). URL: https://beta.gov.wales/sites/default/files/consultations/2018-10/181018-action-on-disability-consultation-v1.pdf

Appendix I: The social determinants of maternal health as cited by the United Nations Development Programme in their Discussion Paper. A Social Determinants Approach to Maternal Health.¹² This figure was adapted from the WHO (2011) Closing the Gap: Policy into Practice on Social Determinants of Health. Geneva: WHO: Page 7.¹¹



This Scientific Impact Paper was produced on behalf of the Royal College of Obstetricians and Gynaecologists by:

Professor GL Jones, Leeds School of Social Sciences, Leeds Beckett University; Dr CA Mitchell DRCOG, Academic Unit of Primary Medical Care, University of Sheffield; Dr J Hirst, Nuffield Department of Women's and Reproductive Health, University of Oxford, John Radcliffe Hospital; and Professor DOC Anumba FRCOG, Academic Unit of Reproductive and Developmental Medicine, University of Sheffield, Sheffield

The following individuals and organisations submitted comments at peer review: Professor S Bewley FRCOG, London; Dr S Bhamani, Aga Khan University, Karachi, Pakistan; Dr P Laxmikant Bhanap, Pimpri Chinchwad Municipal Corporation's Postgraduate institute, Yashwantrao Chavan Memorial Hospital, Pimpri, Pune, India; Ms R Brione, on behalf of Birthrights; Dr F Duxbury, Retired GP, The Leys Health Centre, Greater Leys, Oxford; Professor TP Fleming FRCOG, Southampton; Mr DI Fraser FRCOG, Norwich; Professor M Knight FFPH FRCPE FRCOG, Oxford; Professor A Macfarlane BA Dip Stat C Stat FFPH, London; Dr FS Malik MRCOG, Nottingham; RCOG Racial Equality Taskforce; RCOG Women's Network; ML Treadwell, Research Officer, Birth Trauma Association; Professor TH van den Akker, Leiden, The Netherlands; and Dr IV Walker, University of Southampton, Southampton.

The Scientific Advisory Committee lead reviewer was: Dr RK Morris MRCOG, Birmingham.

The Chair of the Scientific Advisory Committee was: Professor MD Kilby FRCOG, Birmingham.

All RCOG guidance developers are asked to declare any conflicts of interest. A statement summarising any conflicts of interest for this Scientific Impact Paper is available from: <u>https://www.rcoq.org.uk/en/quidelinesresearch-services/quidelines/sipXX/</u>.

The final version is the responsibility of the Scientific Advisory Committee of the RCOG.

The paper will be considered for update 3 years after publication, with an intermediate assessment of the need to update 2 years after publication.

DISCLAIMER

The Royal College of Obstetricians and Gynaecologists produces guidelines as an educational aid to good clinical practice. They present recognised methods and techniques of clinical practice, based on published evidence, for consideration by obstetricians and gynaecologists and other relevant health professionals. The ultimate judgement regarding a particular clinical procedure or treatment plan must be made by the doctor or other attendant in the light of clinical data presented by the patient and the diagnostic and treatment options available.

This means that RCOG Guidelines are unlike protocols or guidelines issued by employers, as they are not intended to be prescriptive directions defining a single course of management. Departure from the local prescriptive protocols or guidelines should be fully documented in the patient's case notes at the time the relevant decision is taken.